CS150 Spring 97

Page 1.1. 1

CS150 Newton 1.1. 1

CS150: Components and Design Techniques

for Digital Systems

Professor A. Richard Newton

566 Cory Hall, 2-2967, rnewton @ ic . eecs

Spring 1997

m Understand Components of Digital Systems

Ü Emphasis on integrated circuit implementations (CMOS)

Ü Especially programmable devices (FPGA, PLD, ROM, RAM, etc.)

m Translation of Design Problems into Hardware Representation

m Design of Hardware Logic Systems

Ü Combinational Logic

Ü Sequential Logic: Synchronous & Asynchronous

m Architecture of Digital Systems

m Arithmetic Circuits and Number Systems

m Introduction to Microprogramming

m Testing of Digital Systems

m Introduction to Analog Interfaces

CS150 Newton 1.1. 2

Organizational Matters

m All handouts will be available on the Web only

Ü http://www- inst .EECS.Berkeley.EDU/~cs150/

Ü Please limit your use of the printers! Try to use them for review

online only.

m Most lectures will be videotaped and tapes are available

for review in McLaughlin Hall.

m All students are required to check the cs150 news

group: cory . eecs . berkeley . edu

within 24 hours of any announcement.

CS150 Spring 97

Page 1.1. 2

CS150 Newton 1.1. 3

Overview

m This Lecture:

Ü What is the course all about & why is it important?

Ü What is a digital system? What is a binary digital system?

Ü Boolean Algebra, Truth tables

Ü Operators:inversion, and,or, xor , xnor ( eq